

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Torm, Nina

Working Paper Does union membership pay off? Evidence from Vietnamese SMEs

WIDER Working Paper, No. 2018/71

Provided in Cooperation with: United Nations University (UNU), World Institute for Development Economics Research (WIDER)

Suggested Citation: Torm, Nina (2018) : Does union membership pay off? Evidence from Vietnamese SMEs, WIDER Working Paper, No. 2018/71, ISBN 978-92-9256-513-8, The United Nations University World Institute for Development Economics Research (UNU-WIDER), Helsinki, https://doi.org/10.35188/UNU-WIDER/2018/513-8

This Version is available at: https://hdl.handle.net/10419/190120

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



WIDER Working Paper 2018/71

Does union membership pay off?

Evidence from Vietnamese SMEs

Nina Torm*

July 2018

United Nations University World Institute for Development Economics Research

wider.unu.edu

Abstract: In the absence of adequate institutional mechanisms, trade unions can potentially promote higher wages and other worker benefits, yet limited data availability means little is known about the effect unions have on individual earnings in developing economies. Using matched employer–employee data from 2013 and 2015 surveys, this paper examines the union wage premium among Vietnamese small and medium-sized enterprises. Controlling for firm and worker characteristics, the results show that unionized workers' wages are 9–22 per cent higher than those of non-union workers. The wage gain is substantially larger at the upper end of the wage distribution.

Keywords: union, wage, small and medium-sized enterprises, Viet Nam JEL classification: C10, J31, J51, M52, O53

Acknowledgements: I would like to thank Finn Tarp, John Rand, and an anonymous referee for helpful comments and constructive feedback. I am also grateful for the productive and stimulating collaboration of the survey teams from the Vietnamese Institute of Labor Science and Social Affairs (ILSSA) and staff at the Central Institute for Economic Management (CIEM). Support from Danida and UNU-WIDER is appreciated.

Copyright © UNU-WIDER 2018

Information and requests: publications@wider.unu.edu

ISSN 1798-7237 ISBN 978-92-9256-513-8

Typescript prepared by Luke Finley.

The Institute is funded through income from an endowment fund with additional contributions to its work programme from Finland, Sweden, and the United Kingdom as well as earmarked contributions for specific projects from a variety of donors.

Katajanokanlaituri 6 B, 00160 Helsinki, Finland

The views expressed in this paper are those of the author(s), and do not necessarily reflect the views of the Institute or the United Nations University, nor the programme/project donors.

^{*} University of Roskilde, Roskilde, Denmark, ninatorm@ruc.dk.

This study has been prepared within the UNU-WIDER project on 'Structural transformation and inclusive growth in Viet Nam'.

The United Nations University World Institute for Development Economics Research provides economic analysis and policy advice with the aim of promoting sustainable and equitable development. The Institute began operations in 1985 in Helsinki, Finland, as the first research and training centre of the United Nations University. Today it is a unique blend of think tank, research institute, and UN agency—providing a range of services from policy advice to governments as well as freely available original research.

1 Introduction

In developed economies, trade unions are generally associated with a positive work environment, including higher wages, increased benefits, reduced wage negotiation costs, and lower wage inequality and worker turnover (Freeman 1980; Standing 1992). In developing and transition countries characterized by different employment structures, including the presence of large informal and unorganized sectors, evidence remains more scarce.¹ However, precisely in such contexts of low wage levels, inadequate institutional support, and limited provision of public sector goods and services, trade unions represent a potentially important collective 'voice' for the promotion of worker benefits and welfare.

In the case of Viet Nam, the transition to a market economy has, along with a growing private sector and the parallel equitization of state-owned enterprises (SOEs), led to the formation of new employment relationships and bases for the establishment of local trade unions. Moreover, in recent years, several changes have been made to the regulatory framework, including the Labour Code and the Trade Union Law, which together provide the impetus for formalizing and strengthening the role of unions. Yet the effectiveness of workplace trade unions remains questionable due partly to their marginal independence, employer dominance, and the generally low incidence of collective agreements.

It is within this context that the current paper analyses whether trade unions are associated with higher wage outcomes, using firm-level survey data on small and medium-sized (non-state) manufacturing firms in Viet Nam from 2013 and 2015 (Rand et al. 2014). The firm panel has been matched with employee data from a subsample of workers in each year. The survey covers micro-, small, and medium-sized firms, and I use the World Bank's classification in terms of the number of employees.² One key advantage of using matched employer-employee data is that it allows for disentangling worker from firm heterogeneity by controlling for firm and worker characteristics which could affect both union status and wage outcomes. Second, in being able to construct a balanced panel of workers some of whom change union status over time, I am able to control for determining factors and selected observed time-varying factors that may simultaneously influence the decision to unionize and subsequent wage outcomes. Third, the availability of numerous firmlevel variables allows for identifying valid instruments so as to capture time-varying unobserved characteristics which, if left unaccounted for, could bias the results. In line with similar studies, the analysis shows a union member wage premium of around 9-22 per cent, depending on the econometric approach. Further robustness checks reveal considerable variation along the conditional wage distribution, with the union wage differential being substantially higher for the top quantiles, in turn calling into question the ability of unions to promote the rights and interests of all workers and their accountability for doing so.

In the following section I provide a selective overview of the existing literature and theoretical background. Section 3 discusses the Vietnamese context, while Section 4 describes the data.

¹ See Eaton et al. (2017) and Schurman and Eaton (2012) for reviews of the growing literature on trade unions and worker organizing in the informal economy.

² The World Bank SME Department operates with three groups of SMEs: micro-, small, and medium-scale firms. Micro-enterprises have between one and ten employees, small-scale enterprises between 11 and 50 employees, and medium-sized enterprises between 51 and 300 employees. These definitions are broadly accepted by the Vietnamese Government (see Government decree no. 90/2001/CP-ND on 'Supporting the Development of Small and Medium Enterprises'). In what follows, I apply these definitions.

Section 5 outlines the methodology and the variables included in the empirical analysis, and Section 6 presents the results of the analysis. The main conclusion, as outlined in Section 7, is that the union-related wage gain is individual, arising directly from being a union member, rather than a spill-over effect from the presence of unions. Moreover, the higher-earning segment of the workforce benefits relatively more from union membership, calling into question the responsiveness of unions to lower-paid workers.

2 Literature and theory

Unions provide workers with a mechanism through which to shape their employment relationship and working environment, by, for instance, improving workplace communication, increasing wages and benefits, and reducing labour turnover (Freeman and Medoff 1984). Thus, through serving as an agent for a firm's employees, a union can take on many different roles and be associated with numerous outcomes, as evidenced by the vast number of trade union studies, covering mostly the US, the UK, and other OECD (Organisation for Economic Co-operation and Development) countries.³ The literature generally portrays unions as operating along at least three dimensions: their wage-making practices; their participatory role in terms of negotiating and administering labour rules related to hiring, training, promoting, and laying off workers; and their activities as a pressure group on government (Pencavel 2005). Rather than attempting to cover the enormous trade union literature as a whole, in this section I address the wage dimension, which is the focus of this paper, by providing a brief theoretical discussion followed by a concise summary of the most relevant studies.

2.1 Theoretical considerations

From a theoretical perspective, the trade union wage gap arises from trade unions introducing a wedge between the reservation wage of the worker and the value of a job, or in other words between labour supply and demand.⁴ The ability of the union to achieve a wage rate that is higher than the non-union level depends on various factors, including the power of the union to act as a monopolist in the supply of labour, and the existence of economic rent or surplus in the product market. In a bargaining model where all union members have the same preferences, wages will be higher the stronger the bargaining power of unions, and the lower the responsiveness of labour demand (and profits) to wages. These relations in turn depend on the competitiveness of the product market. In a perfectly competitive market the union-imposed wedge is reduced, since a firm facing an infinite elasticity of demand is unable to pass any union wage differential on to prices. Thus, unions might be able to capture quasi-rents from capital in the short run, but in the long run the firm would leave the market in search of a higher return on capital, or be forced out by non-union firms with lower costs. In addition to the economic factors, the extent of the union wage gap also depends on the degree of coordination and centralization of wage determination activities and the competitiveness of the labour market (Boeri and van Ours 2008).

In addition to the direct individual union wage premium, spill-over effects to non-union members may occur in the presence of industry-level collective agreements, and in most OECD countries union coverage (the share of workers covered by a collective agreement) exceeds union density (union members as a percentage of the workforce). Since collective agreements ensure that non-

³ See Betcherman (2012) for a review of the literature on labour market institutions in general.

⁴ Booth and Chatterji (1995) and Boeri and van Ours (2008) provide comprehensive reviews of trade union theory. See Lewis (1963, 1986) for earlier work on the union-related wage effect.

members, whether firms or workers, are covered by the union-negotiated wage rate, union coverage is often considered a more accurate measure of the bargaining power of unions than density. The latter has been declining in many OECD countries since the 1980s, and the option to free-ride on union-provided services and benefits is cited as one of the reasons for this (Booth and Chatterji 1995; Boeri and van Ours 2008). In the absence of extension rules, the wages of non-union workers are still likely to be affected by the presence of unions—for instance through the 'threat effect' of unionism, whereby non-union employers will pay wages that are comparable to those of unionized workers (Pencavel 2005). Alternatively, in the absence of such a threat effect (but in the presence of unions), workers disemployed from union firms due to a wage push will seek employment in non-union firms, shifting the labour supply right and reducing wages in this sector below what they would be in the absence of unions. Therefore, whether a threat effect exists or not, wage levels are likely to be affected by the presence of unions.

2.2 Recent empirical evidence

Empirical work on the union–wage relationship has been based mostly on either firm-level *or* individual data, yet since the 1990s a growing number of studies using matched employer–employee data have emerged.⁵ For instance, Lalonde et al. (1996) use American firm data to examine the effects of newly formed unions on total output, employment, material purchases, wage rates, and productivity. They find reductions in the first three, yet firms do not experience higher wage rates. By contrast, in the UK, Hildreth and Pudney (1997) find higher wages among unionized firms, and especially for individuals covered by a collective agreement but who are not necessarily union members. Using matched employer–employee data from Spain, Card and De La Rica (2006) find that firm-level contracting is associated with a 5–10 per cent individual wage premium. Moreover, the wage gain is larger for more highly paid workers, which is in contrast to UK and US studies—where unions have tended to 'flatten' wages across skill groups (Lewis 1986). This difference could, however, be related to the fact that in the UK and US the comparison group is the non-union sector, whereas in Spain it is the prevailing sectoral agreement.

As mentioned earlier, union studies covering the Global South are more limited, and the findings are also more varied. For instance, Rama (2000) states that in developing countries, unionized workers usually earn between 5 and 30 per cent more than non-unionized workers. In the same study, which summarizes findings from Senegal and Cameroon, Rama (2000) shows union membership to be associated with wages that are 8–12 per cent *lower*. This atypical finding is attributed partly to the 'subordinate' nature of the labour movement in these countries and the distortive labour market policies of the 1980s. This work then points to the importance of the nature of the links with the government and political parties when it comes to how labour movements affect wages. In the case of South Africa, Schultz and Mwabu (1998) show that the union wage gap is as high as 145 per cent at the bottom decile of the wage distribution, whereas at the top decile there is flattening so that the difference is 11 per cent. This study, however, is unable to account for important firm attributes, such as size, which are likely to explain much of the union wage gap.

Among the few studies on the union wage premium in Asia, Korean estimates suggest very small positive impacts in the 5–6 per cent range (Fields and Yoo 2000; Park 1991). In a case study of ten Vietnamese firms, Clarke et al. (2007) document that trade unions are able to negotiate wages that are 5 per cent higher than those in non-union firms. Moreover, Edwards and Phan (2008) argue

⁵ See Abowd and Kramarz (1999) for a review of methods and results using matched employer–employee data. In their comprehensive review of studies on developed countries, Aidt and Tzannatos (2002) find the union wage effect to be between 5 and 15 per cent.

that since trade unions in Viet Nam are involved in the central wage decision-making processes, wages would be lower were it not for their presence and influence. However, the study provides no evidence of this.

Methodologically, the above studies are based on either qualitative assessments or cross-sections of individual worker data, and are thus unable to account for workplace characteristics or time-invariant factors. Seeking to fill this gap, an earlier study based on matched firm—worker data from 2007 and 2009 (Torm 2014) found that union membership in Vietnamese small and medium-sized enterprises (SMEs) was associated with higher wages and the provision of social benefits. The current study builds upon and extends this earlier work by constructing a balanced worker panel, allowing for a more comprehensive analysis. Moreover, since 2014 a number of regulatory changes, as examined below, have been implemented, which could influence the results. To set the scene, the next section provides an overview of Vietnamese trade unions and the context in which they operate.

3 Trade unions in the Vietnamese context

Under central planning, trade unions functioned mostly as a 'transmission belt' for the ruling Communist Party, as described in Zhu and Fahey (2000). With Viet Nam's transition to a market economy, a rights-based system for the regulation of industrial relations has been established, allocating a more prominent role to trade unions. As for most developing countries, data on union coverage and membership is generally limited. The Vietnam General Confederation of Labour (VGCL) provides the most reliable and recent statistics, as summarized in Table 1. According to these figures, in 2011 the total number of unions in Viet Nam was just under 110,000, the majority of which were located in the state sector. In the non-state sector, unions were found mostly within domestic enterprises (77 per cent), whereas union incidence among foreign direct investment (FDI) enterprises remains relatively low (13 per cent). In terms of union density, this was 26 per cent across the enterprise sector as a whole in 2011, compared with 16 per cent at the national level (including public servants and armed forces). Total union membership was 7.3 million in 2011, with a slightly higher share in the state sector, yet considering that this sector accounts for two-thirds of unions, the membership ratio appears to be higher in the private sector, as also evidenced by the density figures. More recent figures from 2014 estimate total union membership to be 8.6 million (not reported), thus, assuming a stable wage-earner population, this indicates an increase in union membership over time.⁶ Regarding collective bargaining agreements (CBAs), as indicated in Table 1, national coverage in 2011 was 28 per cent, and was highest in the state sector, at 71 per cent, followed by foreign firms at 54 per cent and domestic firms at 31 per cent. Among unionized establishments the rate was around 67 per cent.⁷

⁶ The VGCL had set a target of recruiting around 600,000 new union members every year between 2013 and 2018 (VGCL 2015).

⁷ To the extent that the figures are comparable this indicates an increase over time, as Clarke et al. (2007) estimated only 20 per cent of unionised private sector firms to have collective agreements (although in Ho Chi Minh City the figure is around 65 per cent).

Trade unions	
Total number of unions	108,302
State sector	76,016
Public agencies	72,009
State-owned enterprises	4,007
Non-state sector	32,286
Non-public agencies	3,338
FDI enterprises	4,060
Domestic enterprises	24,888
Enterprise sector union density	25.8%
National union density (2014)	15.9%
Union membership—total	7,331,832
State sector	3,821,431
Public agencies	2,606,332
State-owned enterprises	1,215,099
Non-state sector	3,510,401
Non-public agencies	79,734
FDI enterprises	1,473,286
Domestic enterprises	1,957,381
CBAs—coverage	
Nationwide	28%
FDI enterprises	54%
Domestic enterprises	31%
State sector	71%
Among unionized establishments	67%

Table 1: Trade union incidence and membership by sector (2011)

Source: author's elaboration based on various VGL data; Chi and Torm (2015).

3.1 The legal framework

The Amended Labor Code (ALC) and the new Union Law (NUL) both became effective in early 2013.⁸ In the ALC chapter XIII, on trade unions, the purpose of firm-based unions is clearly stated as follows: 'trade unions serve to represent and protect the lawful and legitimate rights and interests of trade union members and workers; participate in negotiating, signing, and monitoring the implementation of CBAs, wage scales and wage tables, work norms, wage payment regulations, and bonus regulations among others'. The ALC further stipulates that labour contracts are deemed invalid if, among other circumstances, 'the contract contains an agreement that prohibits or obstructs the employee from forming or joining a trade union'. In addition to protecting the rights and interests of their members, trade union representatives are involved in compliance procedures as stipulated in the NUL, including the implementation of regimes and policies for the employees, such as compliance with minimum wage provisions. The NUL does not, however, set out specific requirements for wages to be above the minimum or average wage in a particular industry, and

⁸ See Decree No. 10/2012/QH13, and No. 12/2012/QH13, for details on the ALC and the NUL respectively.

thus the wage level depends on the bargaining power of the involved parties. However, the employer must consult the executive committee of the enterprise trade union on the formulation of wage scales and labour rates.

The NUL also outlines the functions, tasks, and participation of trade unions in inspecting, supervising, and monitoring the activities of agencies, organizations, and enterprises, and confirms the rights of employees to establish and join trade unions and to participate in trade union activities.⁹ Although enterprise trade unions are free to independently represent workers' rights and interests, all trade unions in Viet Nam are required to affiliate with the VGCL—the higher-level trade union which is the only legal trade union for Vietnamese workers. The VGCL collaborates closely with the local labour department under the supervision of the relevant (Communist) Party body, and this subordination of the trade union to the Party means that workplace trade unions have limited independence and ability to act as a pressure group on government. Moreover, given that trade unions are generally headed by management rather than senior staff members, they are characterized also by weak representational capacity in that dimension.

Compared with the earlier Trade Union Law (1994), the NUL incorporates some major changes, including the requirement for all companies, whether foreign-invested or local, to pay a mandatory union fee of 2 per cent of the total payroll as a social insurance contribution, even if the enterprise has no trade union.¹⁰ The 2 per cent levy is to be used for activities at all levels of trade unions, including the upper-level trade union, which will collect the payment and subsequently distribute part of the fund to the enterprise trade union. Thus, this points to an increased centralization of the ownership of all trade union assets under the VGCL. Another new dimension of the NUL, as also outlined in the ALC, is that the immediate upper-level trade union (usually at the district level) has the right and obligation to represent and protect the legitimate and lawful rights and interests of workers in situations where a grassroots (local) trade union has not been established. Previously, this was only based on employees' requests, with no automatic default protection for workers in non-union firms. In fact, the ALC lays out a number of provisions, including six months' maternity leave, a two-year work permit, and revised working and rest times, which appear to strengthen the position of employees and decrease management flexibility. The impacts of these amendments on employers and the economy as a whole remain to be seen.

3.2 Collective bargaining

The ALC has sought to strengthen collective bargaining by extending the right of workers in nonunionized enterprises to be covered by a relevant collective agreement. The negotiation and monitoring of collective agreements provides an important 'test' of the effectiveness of firm trade unions in representing the interests of their members, and collective agreements should in principle include agreed wage and bonus scales.¹¹ Aside from the fact that one-third of unionized firms do not have such agreements (Table 1), the actual substance of collective agreements is questionable, since they offer few benefits for covered workers beyond conditions provided for by law. For example, a 2009 joint review between the International Labour Organization (ILO) and trade

⁹ Moreover, the NUL outlines the responsibilities owed to trade unions by the state, state agencies, organizations, individuals, and enterprises employing employees; guarantees trade unions' activities; handles dispute resolutions; and deals with breaches of the laws on trade unions.

¹⁰ Other major changes include more protection for part-time trade union officers and prohibiting foreign workers from establishing and joining trade unions and participating in trade union activities.

¹¹ Note that Viet Nam has not yet ratified the international (fundamental) conventions related to freedom of association and collective bargaining set out by the International Labour Organization (ILO).

unions found that most agreements in Viet Nam were initiated by employers to fulfil corporate social responsibility requirements rather than as a consequence of genuine labour–management negotiations (Grimshaw and de Bustillo 2016), demonstrating employers' unwillingness to bargain effectively. Furthermore, collective bargaining rules purposely exclude many categories of workers, such as public sector workers, foreign enterprise workers, subcontracted workers, etc., which is not very conducive to achieving labour market inclusion.

The combination of weaknesses in collective bargaining mechanisms and enterprise unions' weak representation capacity as discussed above have meant that workers have resorted to 'collective bargaining by riots'. Such wildcat strikes, in which workers—without the leadership of union officials—have demanded wages higher than the minimum levels and better working conditions, peaked in 2008 and again in 2011, years during which inflation levels were also high and therefore real wages low.¹² In fact, over 70 per cent of around 5,000 strikes which happened between 1995 and 2011 were on the basis of interests rather than rights only (Chi and Torm 2015). The garment industry, characterized by a combination of low wages and poor working conditions, is the most strike-prone sector in Viet Nam, accounting for 34 per cent of all nationwide strikes recorded during 1995–2014 (VGCL 2015). However, since 2010, when the first sectoral agreement for the garment industry was signed, efforts have been underway to construct the foundations for more effective and resilient sector-level bargaining. There are no official statistics on CBA coverage in the garment industry, yet the Ministry of Labour, Invalids and Social Affairs estimates the CBA coverage to be 50 per cent of unionized companies in 2012 (Chi and Torm 2015). This is below the average coverage rate of 67 per cent among unionized establishments, as shown in Table 1.

The pressure of labour activism has made a growing number of employers adjust their authoritarian approach to wage fixing by allowing for more participation of rank-and-file workers, and their representatives, in *de facto* wage negotiations. Nevertheless, as long as the representation capacity of enterprise unions remains weak, such models of *de facto* wage negotiations do not provide a realistic alternative to collective bargaining. Given this combination of weak union leadership, absence of specific requirements on wage levels, and ineffective collective bargaining, the extent to which unions are associated with higher wages becomes an empirical question. Against this background, I now turn to analysing the union wage premium among Vietnamese SMEs.

4 Data and econometric approach

4.1 Data

The paper is based on matched employer–employee data from two SME surveys, carried out in 2013 and 2015 (Rand et al. 2014). In both years, the surveys included a separate employee module consisting of randomly sampled employees from a random subsample of firms stratified by location. Between one and seven workers were interviewed in each firm, representing different categories. In Viet Nam, household firms are not officially registered by the central authorities under the different enterprise laws (although they are listed by local officials), and as such are not covered by the Trade Union Law.¹³ In other words establishing a trade union is only mandatory

¹² Since then, both inflation and the frequency of strikes have decreased.

¹³ For more detail on coverage see NUL No. 12/2012/QH13, article 3. Moreover, according to Decree No. 88/2006/ND-CP on business registration, when a firm has more than ten employees and/or uses more than one business premise, it may no longer operate as a household firm, and should register as a company under the Enterprise

for firms with more than ten workers and therefore not for household firms. This is evidenced in the data, where only non-household firms report having established firm-level unions and there are no union members observed among household firms. For these reasons, I exclude household firms. and focus solely on firms categorized as private enterprises, cooperatives/collectives/partnerships (CCPs), and limited liability and joint stock companies.¹⁴ After applying this selection criterion and undertaking a thorough data cleaning, including checking the consistency of time-invariant variables between the two survey rounds, I was left with an unbalanced panel of 1,594 permanent workers: 885 in 2013 and 709 in 2015, corresponding to 301 firms.¹⁵ Through identifying repeatedly surveyed employees, I construct a balanced employee panel consisting of 758 individual observations (379 in each year), corresponding to 152 repeated firms.¹⁶

4.2 Union incidence, structure, and benefits

In this section, I consider some descriptive statistics from both the firm and employee survey modules, which could help shed some light on the results in Section 5.

Firm level

Starting with the firms, Table 2, Panel A, shows that 37 per cent of firms have trade unions, and thus the majority of firms do not have local-level unions despite this being mandatory for firms with more than ten workers. However, union density has increased substantially over time, from 33 per cent in 2013 to 43 per cent in 2015, perhaps related to the NUL. The occurrence of local trade unions varies greatly by firm size, from 35 per cent in small firms to 71 per cent in medium-sized firms, and the overall increase in unions over time is attributed to the rise in their occurrence among small firms from around 30 per cent in 2013 to 41 per cent in 2015. Among legal categories, union incidence is more frequent in CCPs and limited liability companies. The latter have also seen a substantial rise over time, as have joint stock companies. Union incidence is least common among private firms, yet for both categories the share has risen considerably over time. As for the different sector categories, shares are higher and relatively even for medium- and high-value-added firms, and all three categories have seen an increase. As will be seen in subsequent sections, union membership varies more for specific sectors.

Law (2000). According to the Amended Labor Code (2012), the application for the registration of Internal Labor Rules (ILRs) which is mandatory for businesses with ten or more employees must include an 'opinion' obtained from the enterprise labour union.

¹⁴ Naturally, firms with fewer than 11 workers (micro-firms) are free to establish unions if they wish, and in the sample 3 per cent of firms with non-household status yet fewer than 11 workers in one of the two survey years indicate having firm-based trade unions. These observations are retained in order to allow for firm dynamics, and to acknowledge that firms which have established unions previously might maintain these despite reducing the number of employees.

¹⁵ I use only worker observations where a monthly wage has been recorded from the outset, rather than recalculating daily/weekly wages into a monthly equivalent, since this relies on certain assumptions about the length of the workday, working week, etc. Moreover, 129 worker observations with missing wages were omitted from the analysis. Results (not reported) from a Heckman selection procedure based on functional form identification reveal no evidence of selection bias along the wage dimension.

¹⁶ The reason for restricting the analysis to two survey rounds is to be able to construct a balanced panel of workers. Since the employee module was not intended to capture repeat workers, adding an additional year would reduce the balanced panel substantially, or even make it impossible to construct one.

Table 2: Trade union incidence by firm characteristics, %

	2013	2015	Mean
Panel A: All firms			
Overall union incidence	0.33	0.43	0.37
Firm size			
Micro	0.10	0.08	0.10
Small	0.30	0.41	0.35
Medium	0.71	0.71	0.71
Legal status			
Private/sole proprietorship	0.27	0.30	0.28
Partnership/collective/cooperative	0.44	0.47	0.45
Limited liability company	0.36	0.46	0.40
Joint stock	0.27	0.46	0.35
Urban	0.40	0.51	0.45
Rural	0.24	0.32	0.28
Sector			
Low value	0.28	0.34	0.30
Medium value	0.36	0.45	0.40
High value	0.30	0.49	0.39
Panel B: Unionized firms			
Unionized workforce	0.80	0.82	0.81
Registered with the VGCL	0.81	0.87	0.84
CBA	0.65	0.67	0.66
Union chair			
Owner	0.02	0.18	0.10
Owner relative	0.02	0.02	0.02
Manager	0.44	0.41	0.42
Head of HR	0.15	0.12	0.14
Senior worker	0.26	0.26	0.26
Other	0.09	0.03	0.05

Notes: Panel A is based on 301 firms (1,594 worker observations). Panel B is based on the 111 unionized firms. Source: Author's elaboration based on SME data (Rand et al. 2014).

Turning now to those firms that have unions, Table 2, Panel B, shows that the membership rate is around 81 per cent and is rising over time. The share of firms that are registered with the higherlevel trade union VGCL (as is mandatory) has also risen from 81 to 87 per cent, which could be related to the NUL. CBA incidence is around 66 per cent, which is very much in accordance with the nationwide 2011 figure (67 per cent) reported in Table 1. The union chair, as discussed earlier, is often a managerial staff member, with 42 per cent of unions being headed by managers compared with 26 per cent by senior workers. Moreover, between 2013 and 2015 there has been a substantial increase in the share of unions chaired by the firm owner. These trends raise doubts about the ability of local trade unions to live up to their mandate of defending labour rights and ensuring social benefits, job security, and safety in the workplace.

Employee level

Turning now to worker-level evidence, drawn from the employee component of the survey, Table 3 shows that union membership increased from 27 per cent to 32 per cent between 2013 and 2015, much in accordance with national union density trends presented in Table 1 (for 2011). Compared with men, a higher share of women are union members, yet over time more men are becoming unionized. Among college graduate workers, union incidence is high at 40 per cent, compared with only 24 per cent of secondary school graduates. Belonging to a union is more common among workers aged 31–45 years, as compared with younger or older workers, but all three categories are quite evenly represented. Service workers have the highest incidence of union membership at 46 per cent, whereas among production workers only around 20 per cent are union members. As illustrated in Figure 1, in terms of the benefits of trade union membership around 37 per cent of respondents replied that 'it secures that the firm pays social benefits', followed by 27 per cent indicating 'more job security', 13 per cent replying 'better and more stable wages', and 11 per cent stating 'more safety at the workplace'.¹⁷ The fact that higher wages are not ranked as the main reason for union membership provides an indication that workers do *not* select into unions in order to obtain higher wages, as discussed in the following section on methodology.

	2013	2015	Mean
All workers	0.27	0.32	0.29
Gender			
Women	0.31	0.34	0.32
Men	0.23	0.30	0.26
Education			
None	0.14	0	0.13
Primary school	0.20	0.17	0.18
Secondary school	0.21	0.28	0.24
College and higher	0.40	0.40	0.40
Age			
15–30 years	0.25	0.32	0.28
31–45 years	0.28	0.33	0.30
> 45 years	0.25	0.29	0.27
Position			
Manager	0.31	0.41	0.35
Professional worker	0.27	0.37	0.31
Office worker	0.36	0.38	0.37
Sales worker	0.33	0.39	0.35
Service worker	0.38	0.58	0.46
Production worker	0.20	0.24	0.22

Table 3: Union membership incidence by employee characteristics, %

Note: Based on 1,594 worker observations.

Source: Author's elaboration based on SME data (Rand et al. 2014).

¹⁷ Torm (2014) finds that union membership is significantly correlated with receiving social benefits, and this holds when restricting the sample to unionised firms only, indicating an absence of spill-overs from union members to non-members when it comes to social benefits.





Source: Author's elaboration based on SME data (Rand et al. 2014).

4.3 Econometric approach

In order to analyse the union–wage relationship, I estimate an equation where individual wages depend on both worker attributes and the characteristics of the firm where the worker is employed. Building on the basic model of Abowd and Kramarz (1999), the specification takes the following form:

$$\ln Y_{ijt} = \alpha + X_{ijt}\beta + Z_{jt}\gamma + U_{ijt}\delta + \varepsilon_{ijt}$$
⁽¹⁾

Where the log of (real) individual wages for worker *i* in firm *j* at time *t* depends on a set of individual worker characteristics (X_{ijt}) and a vector of firm-level covariates for the firm where worker *i* is employed (Z_{jt}) . Our main variable of interest (U_{ijt}) is an indicator for whether the individual is a member of a firm-level trade union. Finally, there is the worker-specific error term (ε_{ijt}) .

In estimating Equation 1, several potential biases are taken into account. First, in order to address the possibility of autocorrelation arising from repeated observations over time, the standard errors are clustered at the firm level. This allows for intragroup (within-firm) correlation over time and between workers, while maintaining the assumption that the observations are independent across firms. Second, bias may arise from the presence of unobserved individual heterogeneity, such as worker ability, which influences both wages and union membership. For example, an observed wage differential between unionized and non-unionized workers may arise simply because the workers who are most likely to become union members are also the ones with the highest unobserved abilities. In order to account for such worker fixed effects (FE), e.g. ability, in the wage specification, we estimate Equation 1 using the balanced panel. Third, bias may arise if changes in union-related policies (for instance as result of the new Trade Union Law which came into effect 1 January 2013) that influence membership are also correlated with changes in wages. To overcome these potential sources of endogeneity bias, I use standard matching techniques to control for determining factors and selected observed time-varying factors that may simultaneously influence the decision to unionize and subsequent wage outcomes. More specifically, I compare differences in wages between workers who unionized in the period 2013-15 and similar workers who remained non-unionized in the 2015 survey.

Finally, I use IV (instrumental variable) identification in order to control for time-varying *unobserved* characteristics, e.g. if the decision to unionize is a function of the perceived wage increase, beyond what is captured by unobserved fixed effects (ability) or observed changes in firm/worker attributes.¹⁸ To instrument for union membership (the endogenous variable), I use the share of firms that have unions, by district and four-digit sector. In order to be able to assess the validity of the instrument (test for overidentifying restrictions), I use a second instrument—the share of firms reporting having good knowledge of the Labour Code, again by district and four-digit sector. The key assumptions of this identification strategy are that (a) the IVs are strongly correlated with the individual union membership, and (b) after adding the full set of worker and firm controls, the selected instruments have no independent influence on individual wage outcomes. If the IV assumptions hold, any observed relationship between union status and wages has a causal interpretation for workers whose union membership status is affected by the instrument (Angrist et al. 1996).

4.4 Descriptive statistics

Table 4 presents the descriptive statistics for the unbalanced panel. First, the main variable of interest is union membership, taking the value 1 if the worker is a member of a union and 0 otherwise. As already seen in Table 3, union membership averages 29 per cent and increased from 27 per cent to 32 per cent between 2013 and 2015. Next, the outcome variable is the logged monthly real basic wage, which in nominal terms increased from 3.7 million Vietnamese dong (VND) in 2013 to VND4.6 million in 2015. The increase in the real wage is less steep, from VND3.2 million in 2013 to VND3.7 million in 2015.¹⁹

In terms of worker attributes, I control for gender, age, and education and training of the worker, as well as job function, hiring method, and reasons for choosing current job. The justifications for the selection of these covariates and their summary statistics are as follows. First, previous studies on Viet Nam (Bjerge et al. 2016; Torm 2014) have shown substantial gender wage gaps, which are also commonly found in other developing country studies (Jones 2001). Thus, I incorporate a gender dummy and Table 4 shows that 55 per cent of the sample are male workers. Second, I control for the age of the worker as a proxy for experience—a key variable in the standard human capital earnings function (Mincer 1974)-and include age squared to allow for a diminishing marginal effect. The average worker age is 36 years. Third, I include a series of education indicator variables, since educational attainment explains a large share of the variation in earnings across individuals (Mincer 1974; Spence 1973). Moreover, this also removes some of the bias arising from the possibility that unionized firms may hire better-quality workers. Around 64 per cent of the sample have finished secondary education, while the share of workers with a higher education is 34 per cent, and this has risen over time. Fourth, the different job functions are included as dummy variables on the basis that both wages and union membership are likely to vary substantially across occupation categories, beyond what is accounted for by education. The largest share is production

¹⁸ Bias arising from self-selection, i.e. the decision to participate in unions being a function of the perceived wage increase, is unlikely, however, since, as indicated earlier, when asked about the main benefit of being a union member, only 13 per cent of workers answered 'better and more stable wages', indicating that workers do not generally select into unions in order to obtain higher wages. Moreover, as indicated in Appendix Table A2, workers who have chosen their current job based on the salary are significantly less likely to be union members.

¹⁹ Real wages are deflated using province-level deflators, where the base is Hanoi, year 2010. In nominal terms, the average wage is in line with the average monthly earnings of salaried workers reported by the General Statistics Office, which were VND3.8 million in 2012 and VND4.5 million in 2014 (GSO 2015). The wages we observe are substantially higher than the current minimum wage, which, depending on the region, was between VND1.65 million and VND2.35 million in 2013 and between VND2.15 million and VND3.10 million in 2015 (Decree No. 103/2014/ND-CP).

workers, at 48 per cent—a share that has increased over time, whereas managers and professional and service workers have seen declining shares between 2013 and 2015.²⁰ Fifth, I include a dummy variable for whether the worker found their job through an informal contact (knowing the owner or someone who works in the firm) as opposed to via a formal contact (advertisement, employment agency, etc.), as this has been shown to be associated with higher individual wages (Larsen et al. 2011). Informal hiring remains the most common recruitment mechanism, at 67 per cent of workers, yet with a slight decrease over time-an indication of increased formalization of the Vietnamese labour market. Sixth, training incidence is added since this has been shown to have an impact on workers' wages in Viet Nam (Bjerge et al. 2016). In addition, theoretical models have predicted that union workers receive more training and higher returns to training than non-union workers.²¹ Seventh, based on the reasoning that job choice preferences may be related to both union status and wages, two indicator variables are included representing whether the worker chose their current job due to working conditions or for salary reasons. Table 4 shows that 23 per cent indicate having selected their job due to better working conditions and 26 per cent due to a higher salary-both have risen over time, indicating that workers are having more influence on the kinds of jobs they attain.

Turning to firm characteristics, I include firm size, legal status, location, sector, wage determination method, owners' gender and education, and the share of professional workers, casual workers, and women. First, firm size (logged) is to account for the commonly found firm-size wage premium (Oi and Idson 1999; Söderborn et al. 2005), and the possibility that larger firms may be more likely to be unionized. The average firm has 38 workers, and this has increased over time. Second, due to the substantial variation in wages and unionization across firm ownership types, I incorporate dummies for the different legal categories. In fact, ownership form has been shown to be a critical factor influencing human resource practices, including in relation to trade unions (Zhu et al. 2008). Limited liability companies comprise the largest category at 60 per cent, followed by private firms at 20 per cent, joint stock companies at 14 per cent, and finally CCPs at 6 per cent. The share of limited liability companies has increased over time, while the share of private firms has fallen, in line also with the overall rise in firm size. Third, I also control for firm location, as wages and human resource initiatives differ across provinces (Zhu et al. 2008), possibly due to the relative autonomy of provinces in the implementation of centrally planned initiatives (Nguyen et al. 2007). Table 1 shows that 58 per cent of firms are located in urban areas (Ho Chi Minh City, Hanoi, and Hai Phong). Fourth, wage returns and human resource practices may vary across sectors of production (Zhu et al. 2008), and thus I control for the different sectors. The majority of firms fall into the medium-value-added category,²² and this is increasing over time. Fifth, around 33 per cent of firms respond that individual wage negotiation, which is likely to be associated with both individual wages and union membership, was the most important determinant of wages.

²⁰ A similar pattern was observed during the 2007–09 period (Torm 2014).

²¹ See for instance Booth and Chatterji (1995).

²² Low-value-added sectors include food and beverages, textiles, apparel, leather, and recycling. Medium-value-added sectors include wood, paper, publishing, rubber, petroleum, chemicals, non-metallic minerals, basic metals, and fabricated metals. High-value-added sectors include electronic machinery, vehicles, transport equipment, and furniture.

Table 4: Summary statistics

	2013		2015		All		
	Mean	SD	Mean	SD	Mean	SD	
Member of a trade union	0.27	0.44	0.32	0.47	0.29	0.45	
Nominal monthly wage VND1,000	3,733.50	1,306.09	4,643.70	2,213.03	4,138.35	1,824.29	
Real monthly wage VND1,000	3,178.79	1,073.89	3,652.00	1,758.40	3,389.27	1,438.58	
Gender (male = 1)	0.55	0.50	0.55	0.50	0.55	0.50	
Worker age	34.14	9.35	38.65	9.32	36.15	9.60	
None	0.01	0.09	0.00	0.04	0.01	0.07	
Primary school	0.01	0.11	0.02	0.13	0.01	0.12	
Secondary school	0.66	0.47	0.62	0.49	0.64	0.48	
College and higher	0.32	0.47	0.36	0.48	0.34	0.47	
Manager	0.12	0.32	0.09	0.28	0.10	0.31	
Professional worker	0.15	0.35	0.13	0.33	0.14	0.35	
Office worker	0.13	0.34	0.14	0.35	0.14	0.34	
Sales worker	0.09	0.29	0.09	0.28	0.09	0.29	
Service worker	0.06	0.23	0.04	0.20	0.05	0.22	
Production worker	0.45	0.50	0.51	0.50	0.48	0.50	
Informally hired	0.69	0.46	0.65	0.48	0.67	0.47	
Training	0.18	0.38	0.27	0.45	0.22	0.41	
Job choice conditions	0.21	0.41	0.26	0.44	0.23	0.42	
Job choice salary	0.24	0.43	0.27	0.45	0.26	0.44	
Firm size	34.22	39.73	41.64	48.66	37.52	44.07	
Private	0.23	0.42	0.17	0.38	0.20	0.40	
Collective	0.06	0.23	0.06	0.24	0.06	0.23	
Limited liability	0.57	0.50	0.63	0.48	0.60	0.49	

Joint stock	0.15	0.35	0.14	0.34	0.14	0.35
Urban	0.57	0.50	0.59	0.49	0.58	0.49
Sector low value added	0.29	0.46	0.27	0.44	0.28	0.45
Sector medium value added	0.43	0.50	0.46	0.50	0.44	0.50
Sector high value added	0.25	0.43	0.25	0.43	0.25	0.43
Wage negotiations	0.32	0.47	0.34	0.47	0.33	0.47
Owner is male	0.50	0.50	0.44	0.50	0.48	0.50
Owner has higher education	0.95	0.22	0.97	0.17	0.96	0.20
Share of women in labour force	0.41	0.22	0.41	0.23	0.41	0.22
Share of professionals	0.10	0.09	0.09	0.10	0.09	0.09
Share of casual workers	0.07	0.15	0.05	0.14	0.06	0.15
Observations	885		709		1,594	

Note: Summary statistics are for the unbalanced panel. Real wages are deflated using province-level deflators. VND20,500 is around US\$1.

Source: Author's elaboration based on SME data (Rand et al. 2014).

Sixth, the gender of the owner has been shown to be important in terms of compensation, with female owners being more generous in the provision of non-wage benefits (Rand and Tarp 2011). Between 2013 and 2015, female-owned firms became more dominant, as indicated by the fall in male-owned firms from 50 to 44 per cent. Seventh, as well-educated managers are more likely to hire well-educated workers (Rosenbaum et al. 1999), I include a dummy indicating whether the owner has at least high school education. The share is very high, at 96 per cent, and increasing over time. Eighth, the share of female workers is included because this is likely to be correlated with lower overall wage outcomes, given the gender wage gap (Croson and Gneezy 2009). Ninth, the share of professional workers acts as a proxy for the general skills level of the workforce, which could be (positively) correlated with unobserved worker-specific ability. Moreover, if, as a result of unionization, firms choose to hire better-quality workers such that productivity increases are commensurate with the union-imposed higher wage (Lewis 1986), average worker quality would eliminates the bias that might arise from a positive correlation between union status and worker quality. Professionals make up around 9 per cent of the workforce, and the share is stable over time. Tenth, the casual workforce share (measured as the average number of casual workers relative to the average number of full-time regular workers in a year) is included as a measure of the stability of employment contracts, as this could be related to both union and wage outcomes. Given that the sample considers only formal firms, it is not surprising to see a relatively low share of casual workers, at around 6 per cent.

Table A1 shows differences in firm and worker characteristics by union membership status. As expected, (raw) wages are higher for union members, and women are more likely to be union members. Union members are slightly older than non-union members, but this is not significant. As also seen in Table 3, union members are well educated and are significantly more likely to have a college education compared with non-union workers, who are more likely to have stopped studying after secondary school. In line with the education figures, managers, office workers, sales workers, and service workers are significantly more likely to be union members (professionals also, but not significantly), whereas among production workers non-unionization is more common. Union members are less likely to have been hired informally, and training incidence is substantially higher among unionized workers-as would be expected given that unions have a longer time horizon and promote the value of training, and as also predicted by theory (Booth and Chatterji 1995). The incidence of workers who chose their current job due to the salary is lower among union members, indicating that higher wages are not a driver of unionization. As expected, union members tend to work in larger limited liability companies (and CCP firms), located in Ho Chi Minh City and Quang Nam, whereas private, more rurally based firms have a higher share of nonunion members. Moreover, union workers are found in firms with highly educated female owners, and a higher share of professional workers. Finally, in terms of sectors, the ratio of union members is significantly higher in wood, chemical products, and fabricated metal products. Appendix Table A2 presents the union membership probit estimations underlying the analysis in Section 5. The results show that differences between union and non-union members clearly exist along the following dimensions: worker age, job function, firm size, CCPs, and female and professional workforce share.²³ Moreover, workers in firms based in Ho Chi Minh City are significantly more likely to be union members, as are those in high-value-added sectors, especially motor vehicles (results not presented). Finally, workers who chose their current job based on the salary are significantly less likely to be union members, providing further support for the argument that higher wages are not a driver of unionization. In the following section, I analyse the main results.

²³ Female-headed firms are more likely to have unions, as indicated by the negative sign on the gender (male owner) coefficient, and the latter is significant if the female workforce share is omitted.

5 Results

In this section I present the overall results, followed by some further robustness checks using quantile regressions, and restricting the sample to only those firms that have unions.

5.1 General

Table 5 shows the relation between union membership and individual wages. In Column 1, when employee attributes only are included, the union wage gap is 13 per cent, yet when the first set of firm-level characteristics (size, legal status, sector, and location) are added in Column 2, the wage premium falls to 10 per cent, although this is still highly significant. In Column 3, the second set of firm-level characteristics are added (owner gender and education, workforce shares, and wage determinant), and the union coefficient remains just over 10 per cent. In order to account for the possibility that unobserved worker heterogeneity is driving the results, I use the balanced panel of workers to control for worker fixed effects (i.e. ability or motivation). Column 4 shows that the union wage gap remains when worker fixed effects are accounted for, yet the size of the coefficient drops slightly, to 9 per cent, indicating that unobserved heterogeneity is *positively* correlated with union membership. This suggests that higher-ability workers are more likely to be unionized. This is in line with Table 3, which showed that among union members the education category that comprises the highest union member share is college education, whereas among non-union workers the share falls as education levels rise. Also as shown in Table 3 and Appendix Table A2, non-production workers are more likely to be in unions than production workers. The small difference in the OLS and FE union coefficient is reassuring and indicates that the numerous controls account for a large share of ability/other unobserved heterogeneity. Assuming that the union membership contribution, which amounts to 2 per cent of wages (and which, as outlined earlier, is now a mandatory fee imposed on the employer), has not been deducted in the reported wage, the real wage gain would be slightly lower. The results are in accordance with Clarke et al. (2007) and Torm (2014).²⁴

²⁴ Similarly, in the case of Spain, Card and De La Rica (2006) showed that unions were associated with a 5–10 per cent individual wage premium.

Table 5: Wages and union membership

	(1)	(2)	(3)	(4)
	OLS	OLS	OLS	FE bal.
Member of a trade union	0.133***	0.101***	0.105***	0.091*
	(0.027)	(0.026)	(0.025)	(0.055)
Gender (male = 1)	0.105***	0.107***	0.097***	
	(0.019)	(0.019)	(0.019)	
Worker age	0.027***	0.024***	0.024***	
	(0.006)	(0.005)	(0.005)	
Age squared	-0.306***	-0.266***	-0.261***	
	(0.078)	(0.066)	(0.066)	
Manager	0.308***	0.304***	0.305***	0.191**
	(0.033)	(0.032)	(0.032)	(0.082)
Professional worker	0.120***	0.121***	0.120***	0.101
	(0.028)	(0.028)	(0.027)	(0.071)
Office worker	0.091***	0.092***	0.089***	-0.007
	(0.028)	(0.028)	(0.028)	(0.055)
Sales worker	0.087***	0.082***	0.082***	0.067
	(0.030)	(0.029)	(0.029)	(0.073)
Service worker	-0.042	-0.054**	-0.062**	-0.067
	(0.028)	(0.027)	(0.027)	(0.080)
Informally hired	-0.023	-0.007	-0.007	-0.038
	(0.022)	(0.025)	(0.025)	(0.037)
Training	-0.016	0.005	0.004	-0.008
	(0.027)	(0.024)	(0.023)	(0.051)
Job choice conditions	0.000	-0.015	-0.020	-0.027
	(0.021)	(0.019)	(0.019)	(0.041)
Job choice salary	0.058***	0.042**	0.032	0.013
	(0.021)	(0.020)	(0.020)	(0.040)
Education controls	Yes	Yes	Yes	Yes
Firm size	Yes	Yes	Yes	Yes
Legal ownership controls	No	Yes	Yes	Yes
Sector controls	No	Yes	Yes	Yes

No	Yes	Yes	No
No	No	Yes	Yes
No	No	Yes	Yes
No	No	Yes	Yes
No	No	Yes	Yes
0.22	0.29	0.30	0.21
			379
1,594	1,594	1,594	758
	No No No No 0.22 1,594	No Yes No No 1,594 1,594	No Yes Yes No No Yes 0.22 0.29 0.30 1,594 1,594 1,594

Notes: OLS are based on the unbalanced panel. Year dummy included in all specifications. Standard errors (in parentheses) are clustered at the firm level. * p<0.10; ** p<0.05; **** p<0.01.

Source: Author's elaboration based on SME data (Rand et al. 2014).

As for the individual worker attributes, these generally conform to human capital theory. First, there is a substantial gender wage gap, with male earnings being around 10 per cent higher than those of women, depending on the exact specification. This is in line with similar studies (Bjerge et al. 2016) and appears to have fallen slightly over time (Liu 2004; Torm 2014). Second, the age of the worker is also highly significant and has the expected concave effect, with a maximum at around 40 years of age. Third, the different occupation categories indicate a substantial wage premium compared with production workers, especially for managers, who earn around 30 per cent more—although this falls to 19 per cent when ability is accounted for in the fixed effects specification. Fourth, as expected, wages are higher if the worker has chosen their job based on the salary. With regard to education (not reported), the results show positive and increasing returns, with a college graduate having a wage that is 19 per cent higher than that of a worker with no education. Yet when the firm-level control variables are added the significance disappears.

For presentational purposes, firm-level variables are not reported, yet larger firms pay significantly higher wages (Söderborn et al. 2005) and individuals working in a CCP have significantly lower wages compared with private-firm employees. Education and gender of the firm owner are not well determined, yet the share of women in the workforce is significantly negative and indicates that shifting from a 0 to a 100 per cent female worker share is associated with approximately a 13 per cent lower wage bill, which is in line with similar studies (Card and De La Rica 2006).²⁵ Both the temporary worker and professional worker shares are insignificant, the latter possibly due to being accounted for by the individual education variables. The location controls (not reported) show that workers in urban areas (Ho Chi Minh City and Hanoi) earn more than those based in rural areas. This is most likely because firms in urban areas pay an efficiency wage in order to attract more productive workers. This is in line with the Vietnam Provincial Competitiveness Index (Malesky 2009), according to which Hanoi and Ho Chi Minh City ranked no. 1 and no. 4 respectively in the 2014 labour policy sub-index, which among other components includes a measure of labour quality (PCI 2015). High-value-added sectors (especially transport equipment and furniture) generally pay higher wages than low-value-added sectors, yet medium-value-added sectors (in particular rubber and refined petroleum) generally pay lower wages than low-valueadded sectors. Further, the year dummy shows an increase in real wages of about 6 per cent between 2013 and 2015.

In Table 6 I zoom in on those workers who became union members between 2013 and 2015, and compare the difference in their wages to that of similar workers who remained non-unionized in 2015. This matching approach is applied to take account of determining factors and selected *observed* time-varying factors that may simultaneously influence the decision to unionize and subsequent wage outcomes. In terms of control variables, Column 1 corresponds to the first column of Table 5, Column 2 to the second, and Column 3 to the third. In Column 1, when only employee characteristics (all lagged) are included, the union coefficient is just below 15 per cent, which is reasonably in line with Table 5. However, once firm controls are added, becoming a union member increases wages by between 22 and 28 per cent, depending on the number and type of firm-level controls. The matched results, however, do not consider the endogeneity bias arising from the possibility of workers self-selecting into unions due to *unobserved* time-varying factors that also simultaneously influence individual wages.²⁶

²⁵ I note that both the gender wage gap and the share of female wage gap have fallen in magnitude by 4 per cent and 12 per cent respectively since Torm (2014), based on the 2007–09 data. This is a positive development, indicating progress in achieving gender equality along the pay dimension, at least.

²⁶ In addition to nearest-neighbour matching I also use double-difference matching, and the results are qualitatively the same, yet the coefficients are smaller in magnitude in all columns (result available upon request).

Table 6: Impact of union membership on wages, matching estimates

	(1)	(2)	(3)
	ATET	ATET	ATET
Becoming union member	0.147**	0.277**	0.222**
	(0.066)	(0.119)	(0.103)
Treated	38	38	38
Observations	275	275	275

Notes: Average treatment effect of the treated (ATET) using bias-corrected nearest-neighbour matching (four matches per observation). Estimations done using the 'teffects nnmatch' command in Stata (Abadie and Imbens 2006, 2011). Matching is based on initial 2013 observed characteristics as documented in the corresponding columns in Table 5. Standard errors (in parentheses) are independent and identically distributed (iid). Matched difference-in-difference (DD) estimates are comparable, yet the coefficients are slightly smaller in magnitude (results available upon request). * p<0.10; ** p<0.05; *** p<0.01.

Source: Author's elaboration based on SME data (Rand et al. 2014).

Thus, in order to deal with potential *unobserved* time-varying factors, in Table 7 I present the instrumental variables estimation (2SLS), using the instruments as described earlier, and the results confirm the positive effect of union membership on wages of workers. Columns 1a and 1b are based on the unbalanced panel, showing a union wage gap of 12 per cent—which, reassuringly, is in line with the earlier results in Tables 5 and 6. As seen in the first-stage estimation in Column 1b, the chosen instruments are significantly related to union membership. In Column 2b, based on the balanced panel, the union coefficient is higher (21 per cent) than the FE in Table 5, but comparable to the matching estimate in Table 6. As reported in the bottom rows of Table 7, the joint test of significance (F-stat) indicates that the instruments are jointly statistically significant at the 1 per cent level. Moreover, the validity of the instruments is confirmed by the Sargan and Basmann tests of overidentifying restrictions (OID), whereby I am unable to reject that the instruments are valid. All in all, this suggest that the chosen instruments influence wages only through their effect on union membership.

U U	•			
	(1a)	(1b)	(2a)	(2b)
	First stage	Wages	First stage	Wages
Member of a trade union		0.116***		0.211***
		(0.035)		(0.044)
Trade union—share of firms	0.706***		0.756***	
	(0.033)		(0.045)	
Labour code knowledge	0.052** (0.026)		0.040 (0.046)	
r2	0.3	80	0.2	29
Observations	1,5	94	75	58
F-stat (instrument relevance)	258.7 0.8	2*** 95	152.8 0.8	89*** 84
OID				

Table 7: Wages	and union	membership.	IV	estimates
Tuble 1. Wayes		membership,	1 V	countaico

Note: 2SLS estimations on the unbalanced sample (Columns 1a and 1b) and the balanced panel (Columns 2a and 2b). Control variables are the same as in Table 5, Column 3. Standard errors (in parentheses) are heteroskedasticity-robust. * p<0.10; ** p<0.05; *** p<0.01.

Source: Author's elaboration based on SME data (Rand et al. 2014).

5.2 Robustness checks

The preceding results are all based on the mean distribution of wages, yet it is plausible that the effects of union membership vary over different parts of the conditional wage distribution, and I

explore this through a set of (semi-parametric) quantile regressions which are presented in Appendix Table A3. Column 3 shows that the union membership wage gap for the median earner is 10 per cent, which is in line with the mean estimate in Table 5, Column 3. Interestingly, the results also reveal that the union membership gain increases considerably as we move up the wage distribution, from a wage premium of 4 per cent for the tenth quantile to as much as 22 per cent for the ninetieth quantile, the latter being almost identical to both the matching estimate (Table 6) and the IV estimate for the balanced panel (Table 7). Apart from the coefficient size difference, the tenth quantile is significant at the 5 per cent level, whereas the higher quantiles are significant at the 1 per cent level.²⁷ Thus, unions in Viet Nam do not seem to be bargaining as effectively on behalf of those at the lower end of the wage distribution, and this bias towards the more skilled segment of the workforce is in part related to the management structure and political subordination of unions, as discussed earlier.²⁸

These results differ from those of previous similar studies (Betcherman 2012). For instance, Schultz and Mwabu (1998) find that in South Africa there are large union effects at the lower part of the wage distribution and that these decrease towards the top. In the case of Ghana, Blunch and Verner (2004) show a union membership premium which is significant only for the tenth quantile. Similarly, in the US, Chamberlain (1994) demonstrates that union membership has a larger effect on the lower quantiles than on the higher quantiles of the conditional distribution of wages, and in the case of the UK public sector, Manquilef-Bächler et al. (2009) find a higher union wage return at the bottom of the distribution among males, while for females the premium is constant across the wage distribution.

The analysis so far compares the wages between union members (in unionized firms) and nonunion members (in union and non-union firms). However, these two groups may not be directly comparable, because the latter have not been faced with the choice of becoming union members, unless they purposely chose to work in a non-unionized firm. Thus, the observed wage difference may simply reflect the differential between being employed in a unionized versus a non-unionized firm, rather than the individual wage gain associated with union membership.²⁹ Since about 20 per cent of workers in unionized firms are not union members, it is possible to test the wage gap within unionized firms only. Appendix Table A4 shows that within unionized firms, the wage gain from being a union member remains significant, and the coefficient of 9.8 per cent is remarkably close to that for the overall sample (Table 5). Thus, what is observed throughout the analysis is the *direct* union membership premium—rather than spill-over effects from there being a union at the firm,

²⁷ This variation along the wage distribution is consistent with splitting the sample into the different occupation categories, which reveals that non-production workers who are union members have a higher wage than non-union members, whereas for production workers the difference is not significant (result available upon request). This in turn is related to production workers presumably having fewer options when choosing which firm to work for, meaning that being a union member may not even be a possibility. By contrast, higher-segment occupation categories have more choice and are more likely to be unionized, as also seen in the summary statistics. Splitting the sample into the different legal categories also reveals that the union wage gap is driven largely by private firms (result available upon request).

²⁸ Splitting the sample into firm-size categories reveals that the results are driven by small firms (5 per cent significance), which is due to the fact that they constitute the majority of firms. As indicated in the summary statistics, 25 per cent of firms in the sample are micro-firms, yet union membership rate among these is rather low, at 3.6 per cent (result available upon request).

²⁹ As discussed earlier, the ALC has sought to extend the right of workers in non-unionized enterprises to be covered by a relevant collective agreement, in which case there would be no union-related wage difference associated with being employed in a union versus a non-union firm.

allowing non-union members to free-ride on union members. This finding is not surprising given the low prevalence of collective agreements and lack of effective bargaining in general.³⁰

6 Conclusion

This paper has examined the union wage premium among Vietnamese manufacturing SMEs using matched employer-employee survey data from 2013 and 2015. In contrast to the situation in many developed countries, union membership is on the rise in Viet Nam, and recent regulatory changes are said to have strengthened the position of employees vis-à-vis management. At the same time, however, the NUL contains elements which imply increased centralization of trade union assets, in turn adding to already limited union autonomy and representative power. These aspects, coupled with the transitional environment in which unions operate, make Viet Nam an interesting case study.

Methodologically, the main contribution of the paper, compared with similar studies, is that the data set allows for the construction of a balanced worker panel, making it possible to account for observed and unobserved time-varying factors which may influence both union status and wage outcomes. The results show that union membership is associated with higher individual earnings when both firm and employee characteristics are controlled for, and that this effect holds when using an IV approach to take account of unobserved time-varying factors. Depending on the analytical approaches adopted, the union wage premium ranges from 9 to 21 per cent, which is in line with the literature.

Further analysis reveals substantial variation along the conditional wage distribution, with the union membership gain being 19 per cent larger for the highest percentile compared with the lowest, which is an opposite finding to those of similar studies. If belonging to a union is more beneficial to workers in higher-level positions, this is concerning for the lower-segment workforce, especially given that non-adherence to minimum wage legislation is not uncommon among Vietnamese SMEs (see Hansen et al. 2016). For this reason, and as also laid out by the NUL, unions should play an imperative role in ensuring that compliance with labour regulations is upheld. Moreover, the fact that the union wage effect appears to be strongest for more-skilled workers, implies a widening of the wage-skill differential, which is somewhat concerning.

Finally, making use of the fact that trade union membership is not compulsory and therefore there are non-union workers within unionized firms, it is revealed that the observed wage difference does indeed reflect a direct individual wage gain associated with union membership, rather than a spill-over effect from being employed in a unionized firm. This result might look different if collective agreements were more effective and widespread, and local trade unions more adequately represented all workers' rights equally. The fact that the incidence of firm-based trade unions is on the rise is positive, yet their ability to act in the interests of their members depends to a large extent on the existence and extension of collective agreements based on effective and inclusive bargaining. In this area, Viet Nam still has work to do.

³⁰ Note that when the sample of unionized firms is restricted to the balanced worker sample, the number of observations drops substantially and the significance disappears. Thus, this result should be interpreted with caution.

References

- Abadie, A., and G.W. Imbens (2006). 'Large Sample Properties of Matching Estimators for Average Treatment Effects'. *Econometrica*, 74: 235–67.
- Abadie, A., and G.W. Imbens (2011). 'Bias-Corrected Matching Estimators for Average Treatment Effects'. *Journal of Business and Economic Statistics*, 29: 1–11.
- Abowd, J.M., and F. Kramarz (1999). 'The Analysis of Labor Markets using Matched Employer– Employee Data'. *Handbook of Labor Economics*, 3(2): 2629–710.
- Aidt, T., and Z. Tzannatos (2002). Unions and Collective Bargaining: Economic Effects in a Global Environment. Washington, DC: World Bank.
- Angrist, J.D., G.W. Imbens, and D.B. Rubin (1996). 'Identification of Causal Effects using Instrumental Variables'. *Journal of the American Statistical Association*, 91(434): 444–55.
- Betcherman, G. (2012). 'Labor Market Institutions: A Review of the Literature'. World Bank Policy Research Working Paper 6276. Washington, DC: World Bank.
- Bjerge, B., N. Torm, and N. Trifkovic (2016). 'Gender Matters: Private Sector Training in Vietnamese SMEs'. Working Paper 2016/149. Helsinki: UNU-WIDER.
- Blunch, N.-H., and D. Verner (2004). 'Asymmetries in the Union Wage Premium in Ghana'. *The World Bank Economic Review*, 18(2): 237–52.
- Boeri, T., and J. van Ours (2008). The Economics of Imperfect Labor Markets. Princeton, NJ, and Oxford: Princeton University Press.
- Booth, A.L., and M. Chatterji (1995). 'Union Membership and Wage Bargaining when Membership Is Not Compulsory'. *The Economic Journal*, 105(429): 345–60.
- Card, D., and S. De La Rica (2006). 'Firm-Level Contracting and the Structure of Wages in Spain'. *Industrial and Labor Relations Review*, 59(4): 573–92.
- Chamberlain, G. (1994). 'Quantile Regression, Censoring and the Structure of Wages'. In C. Sims (ed.), *Advances in Econometrics*. New York: Elsevier.
- Chi, D.Q., and N. Torm (2015). 'Wage Fixing Institutions and Their Impacts in Major Garment-Producing Countries: The Case of Viet Nam'. Mimeo. Geneva: International Labour Organization.
- Clarke, S., C.-H. Lee, and D.Q. Chi (2007). 'From Rights to Interests: The Challenge of Industrial Relations in Vietnam'. *Journal of Industrial Relations*, 49(4): 545–68.
- Croson R., and U. Gneezy (2009). 'Gender Differences in Preferences'. *Journal of Economic Literature*, 47(2): 1–27.
- Eaton, A., S. Schurman, and M. Chen (eds) (2017). *Informal Workers and Collective Action: A Global Perspective*. Ithaca, NY, and London: Cornell University Press.
- Edwards V., and A. Phan (2008). 'Trade Unions in Vietnam: From Socialism to Market Socialism'. In J. Benson and Y. Zhu (eds), *Trade Unions in Asia: An Economic and Sociological Analysis*. New York: Routledge.
- Fields, G.S., and G. Yoo (2000). 'Falling Labor Income Inequality in Korea's Economic Growth: Patterns and Underlying Causes'. *Review of Income and Wealth*, 46(2): 139–59.
- Freeman, R. (1980). 'Unionism and the Dispersion of Wages'. *Industrial and Labor Relations Review*, 34(1): 3–23.

Freeman, R.B., and J.L. Medoff (1984). What Do Unions Do? New York: Basic Books.

- Grimshaw, D., and R. de Bustillo (2016). *Global Comparative Study on Wage Fixing Institutions and Their Impacts in Major Garment Producing Countries.* Geneva: International Labour Organization.
- GSO (General Statistics Office of Vietnam) (2015). 'Report on Labour Force Survey 2014'. Hanoi: GSO.
- Hansen, H., J. Rand, and N. Torm (2016). 'The Impact of Minimum Wage Adjustments on Vietnamese Wage Inequality', 1st edition. Hanoi: International Labour Organization.
- Hildreth, A.K., and S.E. Pudney (1997). 'Employers, Workers and Unions: An Analysis of a Firm Worker Panel with Endogenous Sampling, Attrition and Missing Data'. Discussion Papers in Economics. Leicester: Department of Economics, University of Leicester.
- Jones P. (2001). 'Are Educated Workers Really More Productive?' *Journal of Development Economics*, 64: 57–79.
- Lalonde, R., J.G. Marschke, and K. Troske (1996). 'Using Longitudinal Data on Establishments to Analyse the Effects of Union Organizing Campaigns in the United States'. *Annales d'economie et de statistique*, 41–42: 155–86.
- Larsen A.F., J. Rand, and N. Torm (2011). 'Do Recruitment Ties Affect Wages? An Analysis Using Matched Employer–Employee Data from Vietnam'. Review of Development Economics, 15(3): 541–55.
- Lewis, H.G. (1986). 'Union Relative Wage Effects'. Handbook of Labor Economics, 2: 1139-81.
- Lewis, H.G. (1963). Unionism and Relative Wages in the United States. Chicago: Chicago University Press.
- Liu, A.Y.C. (2004). 'Changes in Gender Wage Gap in Vietnam'. Journal of Comparative Economics, 32(3): 586-96.
- Malesky, E. (2009). The Vietnam Provincial Competitiveness Index: Measuring Economic Governance for Private Sector Development. Hanoi: Vietnam Chamber of Commerce and Industry (VCCI) and Vietnam Competitiveness Initiative (VNCI).
- Manquilef-Bächler, A.A., W. Arulampalam, and J.C. Smith (2009). 'Differences in Decline: Quantile Regression Analysis of Union Wage Differentials in the United Kingdom, 1991– 2003'. IZA Discussion Papers 4138. Bonn: Institute for the Study of Labor.
- Mincer, J. (1974). *Schooling, Experience and Earnings.* New York: National Bureau of Economic Research (NBER) and Columbia University Press.
- Nguyen, B.T., J. Albrecht, S.J. Vroman, and D.M. Westbrook (2007). 'A Quantile Regression Decomposition of Urban–Rural Inequality in Vietnam'. *Journal of Development Economics*, 83(2): 466–90.
- Oi, W., and T. Idson (1999). 'Firm Size and Wages'. In O. Aschenfelter and D. Card (eds), Handbook of Labor Economics, Vol. 3. Amsterdam: Elsevier.
- Park, Y.-B. (1991). 'Union/Nonunion Wage Differentials in the Korean Manufacturing Sector'. International Economic Journal, 5(4): 79–91.
- PCI (2015). 'PCI Data 2015. The Provincial Competitiveness Index'. Available at: http://eng.pcivietnam.org/pci-data-c16.html (accessed 2 July 2018).
- Pencavel, J. (2005). 'Unionism Viewed Internationally'. Journal of Labor Research, 26(1): 65-97.
- Rama, M. (2000). 'Wage Misalignment in CFA Countries: Were Labour Market Policies to Blame?' Journal of African Economies, 9(4): 475–511.

- Rand, J., and F. Tarp (2011). 'Does Gender Influence Fringe Benefits Provision? Evidence from Vietnamese SMEs'. *Feminist Economics*, 17(1): 59–87.
- Rand, J., F. Tarp, and N. Trifkovic (2014). 'Characteristics of the Vietnamese Business Environment: Evidence from a SME Survey in 2013'. Hanoi: Central Institute for Economic Management (CIEM).
- Rosenbaum, J., S.R. DeLuca Miller, and K. Roy (1999). 'Pathways into Work: Short and Long-Term Effects of Personal and Institutional Ties'. *Sociology of Education*, 72: 179–96.
- Schultz, P., and G. Mwabu (1998). 'Labour Unions and the Distribution of Wages and Employment in South Africa'. *Industrial and Labor Relations Review*, 51: 680–703.
- Schurman, S., and A.E. Eaton (2012). 'The Transformation of Work: Challenges and Strategies. Trade Union Organizing in the Informal Economy: A Review of the Literature on Organizing in Africa, Asia, Latin America, North America and Western, Central and Eastern Europe'. Report to the American Center for International Labor Solidarity, Rutgers University. Washington, DC: Solidarity Center
- Spence, M. (1973). 'Job Market Signalling'. Quarterly Journal of Economics, 87(3): 355-74.
- Standing, G. (1992). 'Do Unions Impede or Accelerate Structural Adjustment? Industrial versus Company Unions in an Industrialising Labour Market'. *Cambridge Journal of Economics*, 16: 327– 54.
- Söderbom, M., F. Teal, and A. Wambugu (2005). 'Unobserved Heterogeneity and the Relation between Earnings and Firm Size: Evidence from Two Developing Countries'. *Economics Letters*, 87(2): 153–59.
- Torm, N. (2014). 'The Role of Trade Unions in Vietnam: A Case Study of Small and Medium Enterprises'. *Journal of International Development*, 26(2): 207–21.
- VGCL (Vietnam General Confederation of Labour) (2015). 'VGCL Strike Report'. Presented at the International Conference on Labour Relations and Dispute Resolution, Ton DucThang University, Viet Nam, 26–28 March 2015.
- Zhu, Y., N. Collins, M. Webber, and J. Benson (2008). 'New Forms of Ownership and Human Resource Practices in Vietnam'. *Human Resource Management*, 47(1): 157–75.
- Zhu, Y., and S. Fahey (2000). 'The Challenges and Opportunities for the Trade Union Movement in the Transition Era: Two Socialist Market Economies—China and Vietnam'. Asia Pacific Business Review, special issue on 'Globalization and Labour Market Deregulation: Trade Unions in the Asia Pacific Region', 6(3/4): 282–99.

Appendix

Table A1: Differences in employee and firm characteristics by union status

	No union	Union	Difference (N-U)	t-value
Nominal monthly wage 1,000 VND	3,871.27	4,790.76	-919.49	-9.38***
Real monthly wage 1,000 VND	3,187.21	3,882.86	-695.65	-8.98***
Gender (male = 1)	0.57	0.50	0.08	2.75***
Worker age	35.95	36.63	-0.68	-1.29
None	0.01	0.00	0.00	1.03
Primary school	0.02	0.01	0.01	1.13
Secondary school	0.69	0.52	0.17	6.54***
College and higher	0.28	0.47	-0.18	-7.07***
Manager	0.10	0.13	-0.03	-1.71**
Professional worker	0.13	0.15	-0.01	-0.66
Office worker	0.12	0.17	-0.05	-2.79***
Sales worker	0.08	0.11	-0.03	-1.77**
Service worker	0.04	0.08	-0.04	-3.39***
Production worker	0.53	0.36	0.16	5.99***
Informally hired	0.70	0.60	0.10	3.91***
Training	0.19	0.30	-0.11	-4.65***
Job choice conditions	0.22	0.25	-0.02	-1.05
Job choice salary	0.28	0.21	0.07	2.76***
Firm size	28.59	59.32	-30.72	-13.32***
Private	0.23	0.13	0.10	4.46***
Collective	0.05	0.08	-0.04	-2.83***
Limited liability	0.57	0.67	-0.09	-3.46***
Joint stock	0.15	0.12	0.03	1.64**
Wage negotations	0.33	0.33	-0.00	-0.01
Owner is male	0.51	0.39	0.12	4.50***
Owner has higher education	0.94	1.00	-0.06	-5.44***
Share of women in labour force	0.40	0.45	-0.05	-4.35***
Share of professionals	0.09	0.11	-0.03	-5.18***
Share of casual workers	0.06	0.05	0.01	0.93
Ha Noi	0.22	0.20	0.02	1.00
Phu Tho	0.02	0.01	0.01	1.35
На Тау	0.07	0.05	0.02	1.54
Hai Phong	0.14	0.11	0.03	1.45
Quang Nam	0.06	0.10	-0.04	-2.54**
Nghe An	0.10	0.02	0.08	5.80***
Khanh Hoa	0.13	0.04	0.09	5.48***
Lam Dong	0.05	0.04	0.01	0.51
Ho Chi Minh City	0.18	0.37	-0.19	-8.35***
Long An	0.04	0.07	-0.03	-2.95***
Food and beverages	0.14	0.12	0.02	0.92
Tobacco	0.02	0.01	0.00	0.44
Textiles	0.10	0.04	0.06	3.98***
Apparel	0.01	0.02	-0.00	-0.75
Leather	0.08	0.08	0.00	0.18
Wood	0.03	0.10	-0.07	-5.63***
Paper	0.06	0.03	0.03	2.50**

Publishing and printing	0.00	0.00	0.00	
Refined petroleum	0.02	0.07	-0.05	-5.50***
Chemical products	0.12	0.15	-0.03	-1.73*
Rubber	0.07	0.03	0.04	3.15***
Non-metallic mineral products	0.01	0.02	-0.00	-0.61
Basic metals	0.16	0.13	0.04	1.97**
Fabricated metal products	0.07	0.11	-0.04	-2.74***
Electronic machinery	0.01	0.02	-0.01	-1.58**
Motor vehicles etc.	0.00	0.01	-0.01	-2.52**
Other transport equipment	0.06	0.03	0.03	2.51**
Furniture	0.00	0.00	0.00	0.91
Recycling	0.02	0.03	-0.00	-0.46
Observations	1,594			

Notes: * p<0.10; ** p<0.05; *** p<0.01. Real wages are deflated using province-level deflators.

Source: Author's elaboration based on SME data (Rand et al. 2014).

Table A2: Determinants of union membership

	(1)	(2)
Gender (male = 1)	-0.129	0.018
	(0.160)	(0.179)
Worker age	0.060	0.119*
	(0.063)	(0.068)
Age squared	-0.675	-1.224
	(0.744)	(0.800)
Primary school	0.076	0.767
	(0.756)	(1.007)
Secondary school	0.428	0.953
	(1.113)	(1.388)
College and higher	1.123	1.402
	(1.118)	(1.399)
Manager	0.185	-0.248
	(0.194)	(0.238)
Professional worker	-0.076	-0.229
	(0.215)	(0.248)
Office worker	0.305	0.105
	(0.187)	(0.237)
Sales worker	0.322*	0.037
	(0.194)	(0.253)
Service worker	0.999***	0.561*
	(0.248)	(0.317)
Informally hired	-0.402**	-0.092
	(0.199)	(0.250)
Training	0.610**	0.074
	(0.276)	(0.333)
Job choice conditions	-0.031	-0.077
	(0.220)	(0.281)
Job choice salary	-0.392*	-0.623***
	(0.213)	(0.235)
Firm size		1.426***
		(0.205)
Private		-0.079
		(0.635)

Collective		1.714**
		(0.825)
Limited liability		0.140
		(0.503)
Joint stock		0.000
		(.)
Wage negotations		0.254
		(0.328)
Owner is male		-0.288
		(0.337)
Female share		2.187**
		(0.864)
Professional share		4.067**
		(1.777)
Casual worker share		0.304
		(1.067)
Observations	1,594	1,594

Notes: Marginal effects for the pooled logit estimations. Column 1 includes employee characteristics; Column 2 adds firm characteristics including location, sector, and a year dummy. Standard errors (in parentheses) are clustered at the firm level. * p<0.10; ** p<0.05; *** p<0.01.

Source: Author's elaboration based on SME data (Rand et al. 2014).

Table A3: Wages and union membership, quantile OLS, unbalanced panel

	(1)	(2)	(3)	(4)	(5)
	10th	25th	50th	75th	90th
Member of a trade union	0.040*	0.057***	0.095***	0.170***	0.215***
	(0.016)	(0.017)	(0.017)	(0.020)	(0.020)
Observations	1,594	1,594	1,594	1,594	1,594

Notes: Control variables are the same as in Table 5, Column 3. Robust standard errors (in parentheses). * p<0.10; ** p<0.05; *** p<0.01.

Source: Author's elaboration based on SME data (Rand et al. 2014).

Table A4: Wages and union membership in unionized firms

	(1)	(2)	(3)
	OLS	OLS	OLS
Member of a trade union	0.087**	0.100***	0.098**
	(0.033)	(0.037)	(0.037)
r2	0.25	0.38	0.39
Observations	597	597	597

Notes: Control variables are the same as in Table 5, Columns 1, 2, and 3. Standard errors in parentheses. * p<0.10; ** p<0.05; *** p<0.01.

Source: Author's elaboration based on SME data (Rand et al. 2014).